Author’s response to reviews

Title: Acceptability of the Distress Thermometer and Problem List to community-based telephone cancer helpline operators, and to cancer patients and carers.

Authors:

Karen L Hughes (AnnaHawkes@cancerqld.org.au)
Hilary Sargeant (AnnaHawkes@cancerqld.org.au)
Anna L Hawkes (AnnaHawkes@cancerqld.org.au)

Version: 7 Date: 27 January 2011

Author’s response to reviews:

The following changes have been made to the manuscript:

1. Tracked changes have been removed from the article.
2. Tables have been included in the body of the manuscript.
3. The author’s contributions have been revised to read 'all authors read and approved the final manuscript'.